Author’s response to reviews

Title: Indocyanine green fluorescence imaging to assess bowel perfusion during totally laparoscopic surgery for colon cancer

Authors:

Hao Su (164296114@qq.com)
Hongliang Wu (434803632@qq.com)
Bao Mandula (50625160@qq.com)
Shou Luo (luoshou_smu@163.com)
Xuewei Wang (wangxuewei92@163.com)
Chuanduo Zhao (doraemonfan@163.com)
Qian Liu (384089921@qq.com)
Xishan Wang (932790940@qq.com)
Zhixiang Zhou (zhouzhixiangdoctor@126.com)
Haitao Zhou (zhouhaitao01745@163.com)

Version: 1 Date: 02 Feb 2020

Author’s response to reviews:

Dear editor,

Thank you very much for your suggestions.

1. This study does not prospectively assign patients. Actually, it is a retrospective study. We only use indocyanine green fluorescence imaging in operation when we think it is needed. It aims to retrospectively evaluate the feasibility and safety of intraoperative assessment of bowel perfusion in totally laparoscopic surgery for colon cancer using indocyanine green fluorescence imaging. Therefore, I don’t think this study need trial registration and consort.

3. I have added the consent to participate in the declarations.

Thank you very much again.
Best wishes,

Hao Su